Author's response to reviews

Title: Assessing fear of hypoglycemia in a population-based study among parents of children with type 1 diabetes - psychometric properties of the Hypoglycemia Fear Survey - Parent version

Authors:
Anne Haugstvedt (ahau@hib.no)
Tore Wentzel-Larsen (tore.wentzellarsen@gmail.com)
Morten Aarflot (morten.aarflot@hib.no)
Berit Rokne (berit.rokne@igs.uib.no)
Marit Graue (mgr@hib.no)

Version: 4 Date: 17 December 2014

Author's response to reviews:

The response to the reviewers' comments is included in the uploaded letter to editor.